Summary ~ Week 1, April 2, 2004

Highlights of the RHIC Spin Meeting today. 

The NEWS is encouraging towards possible good pp commissioning, and perhaps even, the  possibility of squeezing in some real physics data from Week 3 onwards.  We should be ready!

Mei Bai:

RHIC Access issues:

1.  Siberian Snakes were wired and tested. OK.  Currents ramped 250A (Operating is ~270A, if I remember correctly)

2.    Hydrogen Jet is in the tunnel

3.    Test ramps in RHIC done and successful

 RHIC Injection Setup:

1. Recovered from access

2. One (blue) RF cavity is off

3. Plan to establish orbits with no snake current today

4.   One snake in blue quenched around 4:00PM recovery in progress

Experimental Magnets:

1. Always need STAR and PHENIX mangnets ON

2. PHOBOS off for machine development

SPIN PATTERN:  (ATTN. ALL SPINNERS!)

Blue Beam: + - + - + - + -

Yellow  B: + + - - + + - -

Monday Access: STARTS AT 7AM, 4/5

Anatoly Zelinsky:   The (Polarized) Sourcer!

Reported ~81% polarization consistently at the end of LINAC 

Increase from ~75 last year to ~81 this year is caused by many small improvements, including possible improvement in the polarimetry.

Haixin Huang:  The AGS (Polarization) Specialist!

1. Reported ~(82 pm 2.3)% polarization at INJECTION consistent with Anatoly Z.'s number at the extraction at LINAC.

2. WARM snake installed last month operates well.

3.  Reported ~50% polarization with 1 x 10^{11} protons/bunch seen over a 2 week period, and a with some  dependence on the polarization on bunch intensity.  More studies in progress.   (all at nominally AGS extraction energies).

--Recall we saw about 5 x 10^{10} p/bunch in the best of last year

--Recall we saw on an average ~40% (?) polarization at the above intensities (range from 35 to 45%)
